Author's response to reviews

Title: Pharmacotherapy of elderly patients in everyday anthroposophic medical practice: a prospective, multicenter observational study

Authors:

Elke Jeschke (ejeschke@havelhoehe.de)
Thomas Ostermann (thomaso@uni-wh.de)
Manuela Tabali (mtabali@havelhoehe.de)
Horst C Vollmar (horst.vollmar@dzne.de)
Matthias Kröz (mkroez@havelhoehe.de)
Angelina Bockelbrink (angelina.bockelbrink@charite.de)
Claudia M Witt (claudia.witt@charite.de)
Stefan N Willich (stefan.willich@charite.de)
Harald Matthes (hmatthes@havelhoehe.de)

Version: 4 Date: 20 July 2010

Author's response to reviews:

Dear Editorial team,
we have modified the manuscript according to your request:
- the list of authors has been modified to fit the document
- all mentioned references were completed
- we remove the visible vertical lines from our tables
- all figures are named consistently in the text
- typography was cross checked again

Thank you once again and

Kind regards

Thomas Ostermann